Author’s response to reviews

Title: A novel biomarker TERTmRNA is applicable for early detection of hepatoma

Authors:

Norimasa Miura (mnmiura@hotmail.com)
Yukio Osaki (yosk@osaka-med.jrc.or.jp)
Miki Nagashima (miki.nagashima@celery.ocn.ne.jp)
Michimori Kohno (kohno@web-sanin.co.jp)
Kensho Yorozu (incepk@yahoo.co.jp)
Kohei Shomori (shomori@med.tottori-u.ac.jp)
Takamasa Kanbe (ikanbe@sanmedia.or.jp)
Kenji Oyama (oyamaknj@hotmail.com)
Yukihiro Kishimoto (kisimoto075@sanin-rosai-h.yonago.tottori.jp)
Shigeo Maruyama (maruyama-clinic@athena.ocn.ne.jp)
Eijiro Noma (syokaki@fukuoka-u.ac.jp)
Yutaka Horie (y-horie@saiseikai-gotsu.jp)
Masatoshi Kudo (m-kudo@med.kindai.ac.jp)
Seigo Sakaguchi (syokaki@fukuoka-u.ac.jp)
Yasuaki Hirooka (yhirooka@med.tottori-u.ac.jp)
Hisao Ito (ito@med.tottori-u.ac.jp)
Hironaka Kawasaki (kawasaki002@sanin-rosai-h.yonago.tottori.jp)
Junichi Hasegawa (hasegawa@med.tottori-u.ac.jp)
Goshi Shiota (gshiota@med.tottori-u.ac.jp)

Version: 3 Date: 2 February 2010

Author’s response to reviews: see over
Dear Dr. Hans Zauner,

Thank you very much for your letter of Feb. 1, 2010, with regard to our manuscript (MS: 7290476972927319) together with the comments from the four reviewers. Herewith, I am sending you a file of our revised manuscript. (Pages 1-32 have been retyped.), after we corrected the minor revisions pointed out by two reviewers as your instruction. Also, our manuscript has been checked by native speaker again. We appreciate your valuable comments very much. And we appreciate Dr. Hans Spangenberg and Dr.Giovanni Tarantino for valuable messages.

Our incorporation of the reviewers’ suggestions is as follows:

**Reviewer: Dr. Patrizia Pontisso**

**Reviewer's report:**

Minor points:

Results section (Pag 9, line 9): the number of the figure (Fig 2) might refer to Figure 5 (or table 1?)

pag 10, line 6 : spelling error (different instead of differently).

**Answer:**

Page9, line 9: We corrected “Figure 2” to “Figure 1”. Because this description that reviewer pointed out is derived form Figure 1.

Page10, line 6: We corrected “differently” to “different”, as pointed out.

**Reviewer: Dr. Michael Grusch**

**Reviewer's report:**

Minor revisions:

1) Page 9, line 13 replace "with only with" by "only with", line 15 replace "tumor number of tumor" with "number of tumors" or "tumor number". "TUNNEL" in Fig. 6
and "Results" section should be "TUNEL" as in the "Materials and Methods" section.

**Answer:**
Page 9, line 13: We replaced "with only with" by "only with" according to the reviewer’s instruction.
Page 9, line 15: We replaced "tumor number of tumor" with "number of tumors"
Fig 6: We corrected “TUNNEL” to “TUNEL”.
Result section: page11, line 20 and page 12, line 1: We corrected “TUNNEL” to “TUNEL”.
We appreciate this review very much.

I believe the manuscript has been improved satisfactorily for publication in BMC Gastroenterology.

Best regards,

Norimasa Miura, M.D., Ph.D
Associate professor.

Division of Pharmacotherapeutics,
Department of Pathophysiological and Therapeutic Science,
Faculty of Medicine, Tottori University
86 Nishicho, Yonago, Tottori 683-8503, Japan
tel: 81-859-34-8014
fax:81-859-34-8140
e-mail: mnmiura@grape.med.tottori-u.ac.jp